Author's response to reviews

Title: What's distressing about having Type 1 diabetes? A qualitative study of young adults' and healthcare professionals' perspectives

Authors:

Myles Balfe (balfe_m@yahoo.com)

Version: 2 Date: 16 November 2012

Author's response to reviews:

Hi,

We've made the amendments that the editorial office requested.

1. Included paragraphs at the end of the manuscript on competing interests and authors' contributions.

2. Included details on ethics approval and consent in the methods section.

3. Included a title page.